Author's response to reviews

Title: Safety of intravenously applied mistletoe extract - results from a phase I dose escalation study in patients with advanced cancer

Authors:

Roman Huber (roman.huber@uniklinik-freiburg.de)
Dietrich Schlodder (dietrich@schlodder.de)
Carola Effertz (ceffertz@helixor.de)
Sabine Rieger (srieger@helixor.de)
Wilfried Tröger (troeger@crdt.de)

Version: 1 Date: 08 Sep 2017

Author’s response to reviews:

Dear Editor,

thank you for accepting our manuscript. The necessary editorial changes have been performed:

- E-mail addresses of all authors have been added on the title page
- Trial registration number has been added to the end of the abstract
- Section "Declarations" has been added according to the formatting guidelines.

I hope the manuscript can now be forwarded to the production team.

With kind regards,

Roman Huber
on behalf of the authors